Author's response to reviews

Title: Delayed Diagnosis of Anterior Subluxation of C7 vertebrae in a Post-Traumatic 35 years old with unusual neurological pattern - a case report

Authors:

Maryam Sanaullah (mariamtariq27@hotmail.com)
Abdul Sattar Mohammad Hashim (2neurospi@cyber.net.pk)
Ayesha Sundus (3ayesha_sundus@hotmail.com)
Sanaullah Bashir (sanaullahbashir@gmail.com)
Maheen Rehman (onlymaheen@hotmail.com)

Version: 4 Date: 4 July 2012

Author's response to reviews: see over
The Editor,

Journal of Medical Case report.

Subject: submission of revised Manuscript bearing MS: 7949464247215951.

Dear Editor,

I would like to submit the following manuscript entitled “Delayed Diagnosis of Anterior Subluxation of C7 vertebrae in a Post-Traumatic 35 years old with unusual neurological pattern - a case report” for evaluation and publication.

I affirm that manuscript is now in accordance with the instructions as directed by deputy editor and editor of journal of medical case report. We have thoroughly gone through the literature and to our knowledge this is the first case of its kind i.e presenting this unusual way despite of having post-traumatic anterior subluxation. We would also like to state that to our knowledge and literature reviewed so far, no case of post-traumatic subluxation of cervical vertebrae has ever presented like this. I have read the manuscript and I hereby affirm that the content of this manuscript has not been published in any journal and it is not being submitted fully or partially for publication elsewhere. The manuscript has been read and approved by all the listed authors.

Regards,

Maryam Sanaullah

14-5-2012